Author's response to reviews

Title: Psychotic and schizotypal symptoms in non-psychotic patients with obsessive-compulsive disorder

Authors:

Stian Solem (stian.solem@svt.ntnu.no)
Kristen Hagen (kristeha@stud.ntnu.no)
Christoffer Wenaas (c.wenaas@gmail.com)
Åshild T Håland (aashild.haaland@sshf.no)
Gunvor Launes (gunvor.launes@sshf.no)
Patrick A Vogel (patrickv@svt.ntnu.no)
Bjarne Hansen (angstsenteret@gmail.com)
Joseph A Himle (himlej@med.umich.edu)

Version: 5 Date: 30 April 2015

Author's response to reviews: see over
April 30, 2015

Editor
BMC Psychiatry

Dear Editor

We are very thankful for a helpful and efficient review process. Below you will find our response to the editorial requests.

Sincerely,

Stian Solem
Norwegian University of Science and Technology
Department of psychology
7491 Trondheim
Norway

+ 47 73 55 04 93 (voice)
stian.solem@svt.ntnu.no
Reviewer #1:

1. No further comments. The authors have done a nice job of addressing all previous comments.

   Reply: We would like to thank reviewer #1 for improving our manuscript.

Editorial requests:

1. Please add all authors' email addresses in the title page.
   Reply: All authors’ email addresses are now listed in the title page.

2. Please add a 'Background' title after the abstract for your Background section.
   Reply: We have now included “background” as a title in the introduction section.

3. Please provide the full name of the ethics committee in your Methods section.
   Reply: The full name of the ethics committee is: Regional Committees for Medical and Health Research Ethics. This review board covers all health related research in Norway. We now list the full name in the methods section.

4. Consent statement
   Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian.
   Reply: We now mention that written informed consent was obtained from all participants.

5. Please move Competing Interest, Authors' Contribution and Acknowledgements before the references section.
   Reply: We have moved the tables to the end of the manuscript according to the guidelines of the journal. Competing interests, authors’ contributions and acknowledgements are followed by the references section.